Author’s response to reviews

Title: Surgical Configurations of the Pectoralis Major flap for Reconstruction of Sternoclavicular Defects: A Systematic Review and New Classification of Described Techniques

Authors:

Jude Opoku-Agyeman (jo7845@pcom.edu)
David Matera (davidmate@pcom.edu)
Jamee Simone (js252717@pcom.edu)

Version: 3 Date: 03 Sep 2019

Author’s response to reviews:

Author’s response to reviews

Title: Surgical Configurations of the Pectoralis Major flap for Reconstruction of Sternoclavicular Defects: A Systematic Review and New Classification of Described Techniques

Authors: Jude Opoku-Agyeman (jo7845@pcom.edu), David Matera, Jamee Simone

Author’s response to reviews:

Dear Editor, thank you very much for the time and effort you've put into this manuscript. We would also like to thank the reviewers for their suggestions and time spent to analyze this manuscript.

We have carefully reviewed the comments and have revised the manuscript accordingly.

Response to reviewers and editor

1. Overlap

Answer: The result section has been re-phrased to minimize overlap

2. Figure attribution

Answer: All images in Figure 2 are the author’s own images. The images were hand drawn. Images were captures and loaded into Microsoft photo editor for image manipulation. The
images were then annotated on Microsoft word, cropped using a snipping tool and saves as a JPEG file.

3. Abbreviations
Answer: A list of abbreviations used in the manuscript have been included just before the declaration section

4. Cite
Answer: All figure (Fig.1, Fig.2 and Fig.3) and tables (Table 1.) have been cited within the text

5. PRISMA
Answer: The PRISMA check list has been removed

6. Clean manuscript
Answer: A Clean manuscript has been uploaded. All tables and figure have been loaded separately.

I hope that the changes we have made could resolve your concerns about the article. Thank you again for your time.

Jude Opoku-Agyeman, GME office, Philadelphia College of Osteopathic Medicine, 4190 city avenue, Philadelphia PA, 19131. Phone: +1(515)-864-9619, email: jo7845@pcom.edu